Author's response to reviews

Title: Acupuncture, chiropractic and osteopathy use in Australia: a national population study

Authors:

Charlie CL Xue (charlie.xue@rmit.edu.au)
Anthony L Zhang (tony.zhang@rmit.edu.au)
Vivian Lin (v.liln@latrobe.edu.au)
Ray Myers (Ray.myers@rmit.edu.au)
Barbara Polus (Barbara.polus@rmit.edu.au)
David F Story (David.story@rmit.edu.au)

Version: 3 Date: 11 March 2008

Author’s response to reviews:

11 March 2008

Dr Annabel Phillips
Senior Editor
BMC-Series Journals
BioMed Central
BMC Public Health

Dear Dr. Phillips,

Re: BMC Public Health MS 3391402841634904 ¿Acupuncture, chiropractic and osteopathy use in Australia: a national population-based study¿

Thank you for advising us that our manuscript is accepted for publication in the BMC Public Health. We have checked the contents and the format of the manuscript carefully. Two minor editorial corrections have been made. The revised MS has been uploaded through the submission system.

We look forward to your further advice.

David Story and co-authors